Author's response to reviews

Title: Lack of adherence to hypertension treatment guidelines among GP's in southern Sweden - a case report based survey

Authors:

Rickard W Ekesbo (rickard.ekesbo@skane.se)
Patrik Midlöv (patrik.midlov@med.luse)
Sofia Gerward (Sofia.gerward@med.lu.se)
Kristin Persson (perssonkristin@hotmail.com)
Christina Nerbrand (christina.nerbrand@med.lu.se)
Lennart Johansson (lennartkarin@telia.com)

Version: 4 Date: 11 January 2012

Author's response to reviews: see over
Dalby, Jan 11, 2012

The Editor

BMC Family practice

Dear Sir,

We thank you for the opportunity to resubmit our paper and for the clear views of the Associate editor. Particularly, we appreciate your taking the time for suggestions for clarifications.

We have made an earnest effort to adjust and rewrite the manuscript accordingly, especially noting the disposition of the discussion which also has been made more comprehensive. In addition, we have made some linguistic adjustments as requested.

Again, thanking you for your generous input and with these changes we hope you find the manuscript has improved sufficiently for publication, we remain

Yours sincerely,

Rickard Ekesbo
for all authors